Author's response to reviews

Title: Magnesium administration provokes motor unit survival, after sciatic nerve injury in neonatal rats

Authors:

Nicolaos Gougoulias (gougnik@yahoo.com)
Athanasiou Hatzisotiriou (Hatzitha@yahoo.com)
Dorothea Kapoukranidou (dorodemo@med.auth.gr)
Maria Albani (albani@med.auth.gr)

Version: 4 Date: 15 September 2004

Author’s response to reviews: see over
Dear Ms Iratxe Puebla,

Please find attached all the following formatting changes requested are done. Thank you very much for your consideration,
Best Wishes,
Maria Albani

Formatting changes requested:
----------------------------------------

Title - Please remove all unnecessary capitalisation, i.e. 'MAGNESIUM ADMINISTRATION' should be 'Magnesium administration...'.
Qualifications - Please remove the authors' qualifications (e.g. MD) from the title page.
Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.
Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: [http://www.biomedcentral.com/info/ifora/cropping](http://www.biomedcentral.com/info/ifora/cropping).
Tables- Table 4 has been submitted in two parts. Can we ask you to reformat this table so that all the data is incorporated into a single table labelled Table 4? If this is not possible, please label Table 4a as table 4 and table 4b as table 5. If you do so, please remember to update the citation on the text accordingly.